Author's response to reviews

Title: Non-medical prescribing in Australasia and the UK: the case of podiatry

Authors:

Alan M Borthwick (ab12@soton.ac.uk)
Anthony J Short (anthony@podiatrypractice.com.au)
Susan A Nancarrow (S.Nancarrow@sheffield.ac.uk)
Rosalie Boyce (rosalie.boyce@gmail.com)

Version: 4 Date: 15 December 2009

Author's response to reviews:

The requested changes have been performed as follows;

1. Table has been reformatted and placed within the manuscript following the References
2. References with non-functioning URL links have been edited by removal of the link and additional source details included
3. Author names have been correctly matched to the submission

Many thanks,

Anthony Short